Author's response to reviews

Title: Assessment of urinary kidney injury molecule-1 and interleukin-18 in the early post-burn period for predicting acute kidney injury in patients with various degrees burn injury

Authors:

Hongqi Ren (sznk2005@163.com)
Xuan Zhou (2011snkzx@sina.cn)
Deshu Dai (daideshu0608@126.com)
Xiang Liu (995780980@qq.com)
Liangxi Wang (wlx73@163.com)
Yifang Zhou (xzldxxh@sina.com)
Xaomei Luo (XaomeiLuo@163.com)
Qing Cai (qingdan0000@163.com)

Version: 5 Date: 18 September 2014

Author's response to reviews: see over
Dear Editor:

The manuscript entitled “Clinical significance of urinary kidney injury molecule-1 and interleukin-18 levels for the early diagnosis of acute kidney injury in patients with varying degrees of burns” that we have submitted to in BMC nephrology. MS: 6526114941288376.

According to reviewer's report, we have made a number of revisions. In the first, we have changed the title to “Assessment of urinary kidney injury molecule-1 and interleukin-18 in the early post-burn period for predicting acute kidney injury in patients with various degrees burn injury”, and we re-organize and analyze the data of burn patients.

According to Editorial Request, we have revised my manuscript to include line and page numbers. In addition, we have listed the abbreviation in the end of our paper.

Now I send the revised style to you and I expect it can be proceed with peer review as soon as possible.

Sincerely,
Hongqi Ren, MD, PHD
Department of Nephrology,
Nanjing Clinical College of Second Military Medical University,
305 Zhongshan East Road,
Nanjing, China